Author’s response to reviews

Title: Dietary zinc intake and its determinants among Ethiopian children 6-35 months of age

Authors:
Girmay Mersha (girmayayana@yahoo.com)
Tibebu Moges (tibebumoges@gmail.com)
Aregash Samuel (aregashhafebo@gmail.com)
Tsehai Assefa (tsehiaassefa@yahoo.com)
Solomon Eshetu (Solomoneshetu13@yahoo.com)
Aweke Kebede (awekekeb@gmail.com)

Version: 4 Date: 19 Jun 2018

Author’s response to reviews:
Thank you for considering our manuscript for publication.
The authors had read and approved it.
We arrange the tables and figures according to the journal guideline.
We made all corrections according to the comments and final version is attached.